EECSE 461 Homework 4 Winter 1997 Prof. Tom Anderson Due: Feb. 13, 3:30pm

All questions are equally weighted so make sure you get the easy ones right. You are free to discuss this assignment with other students and the TA, but you must each individually write up and submit your own solutions.

1. Peterson and Davie, Ch. 3, Problem 8 2. Peterson and Davie, Ch. 4, Problem 2 3. Peterson and Davie, Ch. 4, Problem 3 4. Peterson and Davie, Ch. 5, Problem 1 5. Peterson and Davie, Ch. 5, Problem 5 6. Peterson and Davie, Ch. 5, Problem 7 7. Peterson and Davie, Ch. 5, Problem 10 8. Peterson and Davie, Ch. 5, Problem 13